Author's response to reviews

Title: Long term evolution of Valsalva Retinopathy: a case series

Authors:

   Miriam Garcia Fernandez (migarci@hotmail.es)
   Joaquín Castro Navarro (jcastro22557@gmail.com)
   Carmen Gonzalez Castaño (cgcastano@hotmail.com)

Version: 3 Date: 31 May 2012

Author's response to reviews: see over
RESPONSES TO THE REVIEWERS

Regarding formatting changes:

- We have included the study design in our title: a case series.
- We have divided the case presentation in the abstract and in the main manuscript into the six patients presented in this case series. We have added ethnicity, sex and age in each of the patients.
- We have removed the title after the abstract.
- We have specified the amount of time between the cause of the hemorrhage and the presentation of the case report.

Referee 1:

- We have included at the end of the introduction the reason to report these cases: *We report the cases of six patients suffering from premacular hemorrhage secondary to different conditions in order to present the therapeutic procedures used and their long-term outcomes together with a literature review on this topic.*
- We have explained why Nd:YAG laser was performed in two patients.
Referee 2:

-In discussion, we have highlighted the downsides of YAG:laser, enhancing the difficulty in doing the laser so close to the retinal surface.

Many thanks,

The authors.